Author's response to reviews

Title: Does the aldosterone: renin ratio predict the efficacy of spironolactone over bendroflumethiazide in hypertension? A Clinical trial protocol for RENALDO (RENin-ALDOsterone) study

Authors:

Hari K Parthasarathy (krishnankala@hotmail.com)
Khamis Alhashmi (alhashmi1@yahoo.com)
Allan D Struthers (a.d.struthers@dundee.ac.uk)
John MC Connell (mcc1m@clinmed.gla.ac.uk)
Gordon T McInnes (g.t.mcinnes@clinmed.gla.ac.uk)
Ian Ford (ian@stats.gla.ac.uk)
Thomas M MacDonald (t.m.macdonald@dundee.ac.uk)

Version: 2 Date: 13 April 2007

Author's response to reviews:

Trial registration: Current Controlled Trials ISRCTN93126600